Author's response to reviews

Title: Takotsubo cardiomyopathy in an adolescent affected by pancreatic adenocarcinoma

Authors:

Maria Debora De Pasquale (mdebora.depasquale@opbg.net)
Angela Mastronuzzi (angela.mastronuzzi@opbg.net)
Luigi De Sio (desio@opbg.net)
Annalisa Serra (annalisa.serra@opbg.net)
Chiara Grimaldi (chiara.grimaldi@opbg.net)
Ugo Giordano (ugo.giordano@opbg.net)

Version: 2 Date: 2 April 2015

Author's response to reviews: see over
To: Editor-in-Chief, BMC Pediatrics

Title: TAKOTSUBO CARDIOMYOPATHY IN AN ADOLESCENT AFFECTED BY PANCREATIC ADENOCARCINOMA

Authors: De Pasquale Maria Debora,

Dear Editor,

I am pleased to submit the manuscript entitled “TAKOTSUBO CARDIOMYOPATHY IN AN ADOLESCENT AFFECTED BY PANCREATIC ADENOCARCINOMA” for consideration for publication on BMC Pediatrics.

Takotsubo cardiomyopathy in an adolescent has never been reported in medical literature. In this manuscript, we describe a case of an adolescent affected by pancreatic cancer who developed reversible cardiomyopathy during treatment with 5-fluorouracil.

We believe that this manuscript is appropriate for publication by BMC Pediatrics. This manuscript has not been published and is not under consideration for publication elsewhere. We have no conflicts of interest to disclose. All the Authors have been revised and approved the final version submitted to BMC Pediatrics.

In the revised version, as requested:

- I have included a “Consent” section after the conclusion
- I have included line and page numbers in the main manuscript

Thank you for your consideration!

Sincerely,

Best regards.

Maria Debora De Pasquale, MD, PhD